Author's response to reviews

Title: Health care consumption and costs due to foot and ankle injuries in the Netherlands, 1986-2010

Authors:

A. Siebe De Boer (s.deboer@erasmusmc.nl)
Tim Schepers (t.schepers@erasmusmc.nl)
Martien JM Panneman (m.panneman@veiligheid.nl)
Ed F Van Beeck (e.vanbeeck@erasmusmc.nl)
Esther MM Van Lieshout (e.vanlieshout@erasmusmc.nl)

Version: 5 Date: 12 March 2014

Author's response to reviews:

Response to comments of Associate Editor

1. Point 8: "incidence rate for hospital admission": "incidence rate of admitted patients" has consistently been replaced with "incidence rate for hospital admission"

2. Point 10: please change "....was based" to "....were based": the typo was corrected (Methods section - Data sources subsection, last paragraph).